Author's response to reviews

Title: At the coalface and the cutting edge: A qualitative study of the pathways into and rewards of providing HIV medical care in general practice settings

Authors:

Christy E Newman (c.newman@unsw.edu.au)
Asha Persson (a.persson@unsw.edu.au)
John BF de Wit (j.dewit@unsw.edu.au)
Robert H Reynolds (robert.reynolds@mq.edu.au)
Peter G Canavan (petercanavan@me.com)
Susan C Kippax (s.kippax@unsw.edu.au)
Michael R Kidd (michael.kidd@flinders.edu.au)

Version: 2 Date: 28 September 2012

Author's response to reviews:

Apologies for uploading a blinded copy originally, this version has all of the required text including ethics, competing interests, acknowledgements, authors contributions, and authors information.